Author's response to reviews

Title: Job satisfaction trends during nurses' early career

Authors:

Trevor Murrells (trevor.murrells@kcl.ac.uk)
Sarah Robinson (sarah.robinson@kcl.ac.uk)
Peter Griffiths (peter.griffiths@kcl.ac.uk)

Version: 4 Date: 14 May 2008

Author's response to reviews:

To: the Editor-in-Chief

Dear Dr Norton,

We have added in some detail on informed consent and indicated where Tables 1 and 2 should be located.

Regards

Trevor Murrells
National Nursing Research Unit
King's College London